Author's response to reviews

Title: Equal antipyretic efficacy of oral and rectal acetaminophen: a randomized controlled trial [ISRCTN 11886401]

Authors:

Mona MN Nabulsi (mn04@aub.edu.lb)
Hala HT Tamim (ht02@aub.edu.lb)
Ramzi RS Sabra (rsabra@aub.edu.lb)
Ziyad ZM Mahfoud (zm15@aub.edu.lb)
Shadi SM Malaeb (smalaeb@wihri.org)
Hadi HF Fakih (alfakihdr@hotmail.com)
Mohammad MM Mikati (mamikati@aub.edu.lb)

Version: 2 Date: 11 March 2005

Author's response to reviews:

An ISRCTN has been added to the title as shown below:
Equal antipyretic efficacy of oral and rectal acetaminophen: a randomized controlled trial [ISRCTN 11886401]